# ISSN 2282-8168

# CEFIN Working Papers No 80

# Birds of a feather flock together and get money from the crowd

Valeria Venturelli, Giovanni Gallo and Alessia Pedrazzoli

December 2019

# Birds of a feather flock together and get money from the crowd

# Valeria Venturelli\* Giovanni Gallo† Alessia Pedrazzoli‡

#### December 2019

#### **Abstract**

In constructing online alternative finance instruments as a new form of financial democratization and financial inclusion, this article aims at verifying the presence of similarity effect in equity crowdfunding investments. Discussion focuses on ethnic and gender similarity between the seekers and investors that sustained the project. Our analysis is based on 5,996 personal investors that have participated in 81 equity crowdfunding campaigns, on Crowdcube, a British equity crowdfunding platform from 2011 and 2016.

Results show that in equity crowdfunding gender and ethnic similarities play different role based on investors' characteristics - gender, ethnicity and the combination of two. In particular, ethnic similarity positively influence the level of amount invested by both female and male investors belonging to an ethnic minority. Even if female investors tend to prefer male company, their preference changes if a female proponent belonging to an ethnic minority runs the company.

From a practical perspective, our findings shed new light on how individual characteristics can be important factor in financing situations. Results allow entrepreneurs and equity crowdfunding platforms to understand better potential investor behaviour and highlights the role of equity crowdfunding as tool for minorities' financial inclusion and women entrepreneur empowerment.

**Keywords:** equity crowdfunding, entrepreneurial finance, ethnicity, gender, similarity effect

**JEL classifications: G02 G11 G41 M13** 

\_

<sup>\*</sup> Department of Economics Marco Biagi - University of Modena and Reggio Emilia and CEFIN - Centro Studi Banca e Finanza at the University of Modena and Reggio Emilia. Email: valeria.venturelli@unimore.it

<sup>&</sup>lt;sup>†</sup> National Institute for the Analysis of Public Policy (INAPP) and Department of Economics Marco Biagi. Email: giovanni.gallo@unimore.it

<sup>&</sup>lt;sup>‡</sup> Department of Economics Marco Biagi - University of Modena and Reggio Emilia. Email: alessia.pedrazzoli@unimore.it

#### 1 Introduction

Equity crowdfunding has been a rapidly growing financial instrument since 2012 when the financial crisis have worsened the problem of fundraising for small and medium size enterprises (SMEs) and innovative firms creating room for the development of new financial tools and intermediaries.

Compared to other forms of crowdfunding - donation-based, lending based and reward-based - equity crowdfunding is still in its infancy but due to the rapid growth of the market, regulators and policy makers from different countries are paying a great attention on this relatively new phenomenon. Thanks to the web context in which crowdfunding is developed, crowdfunding is expected to democratize funding by increasing the availability of capital to traditionally underrepresented groups and by enabling entrepreneurs to enlarge their appeal for investments (Barasinska and Schäfer, 2014).

Currently, studies on equity crowdfunding especially focus on the characteristics of the seekers side detecting which drivers influence campaign success (Ahlers et al., 2015; Lukkarinen et al., 2016; Vismara 2016a; Vismara, 2016b). From the supply side, much of the prior scholarly research about investors concentrated effort in ascertaining motivation to participate in crowdfunding campaign (Gerber et al., 2012). In reward and donation based platforms, in particular, investors are driven to fund projects by intrinsic motives, such as social reputation, shared identity, or other non-pecuniary benefits, whereas those on equity-based platforms are driven predominantly by extrinsic — financial — motivation (Schwienbacher and Larralde, 2010; Ordanini et al., 2011; Cholakova and Clarysse, 2015) or a combination of both extrinsic and intrinsic drivers (Collins and Pierrakis, 2012). Little is known about investor characteristics, from company and platform perspectives, a better understanding of them reduces the ex-ante search costs and facilitates the ex-post matching (Polzin et al., 2018). For this reason, the need arises of conducting research on the behaviour and individual characteristics of equity crowdfunding investors.

As for investor individual characteristics few studies have addressed gender issues in crowdfunding. In a broad sense, from seekers side, women are less likely than men to engage in new ventures but from the supply side is important to understand if also in crowdfunding market a gender gap persists (Leitch et al., 2018). Compared to reward based crowdfunding where Marom et al. (2014) find that 44% of investors on Kickstarter are female, in investment based crowdfunding, equity and real estate, Hervé et al. (2016) show that 93% of equity crowdfunders are men, so crowdfunding investment market is not different from other

financial markets like business angels and VC that are predominantly male-dominated (Brush et al., 2004, Harrison and Mason, 2007).

Even if the equity crowd and entrepreneurs are mainly male, female entrepreneur are more likely to succeed at crowdfunding campaigns thanks to the support of other female backers (Greenberg and Mollick, 2015; Horvat and Papamarkou, 2017). In reward-based and equity-based model Greenberg and Mollick (2015) and Vismara (2016c) refer to homophily theory and their researches show that there is a strong correlation between female gender of the entrepreneur and backers' profile.

Also ethnic profile is one of the element that may improve similarity effect and it may be crucial in generating trust and overcoming information asymmetry between investors and proponents. To the best of our knowledge, there are no study that try to combine gender and ethnic similarity in attracting investors on equity crowdfunding platforms. Our research aims at verify if the similarity effect increases the amount invested in the campaign by the single investor and different investor's characteristics, also highlighting the relative role of similarity respect company and entrepreneur characteristics. Toward this end, we collected shareholders information for 81 equity crowdfunding campaigns posted on Crowdcube, a British Platform, from 2011 to 2016.

We collected firms information about the campaign from the Crowdcube website while shareholders information - name and amount invested - from documents filling hosted on Companies House website (*companieshouse.gov.uk*). We identify investors' gender we use the *genderize.io* tool while ethnical origin by surname geographic distribution detached through *forebears.io*, a genealogical source.

Our data show that in equity crowdfunding gender and ethnic similarities play different role on the basis of investors characteristics - gender, ethnicity and the combination of two. In particular, ethnic similarity positively influence the level of amount invested by both female and male investors belonging to an ethnic minority. Even if female investors tend to prefer male company, their preference changes if the company is run by a female proponent belonging to an ethnic minority. Among the aspect that influence the variability of the total amount invested, entrepreneur profile and company aspects are the most effective dimensions; nonetheless, also gender and ethnic similarity play a role in determine the total amount invested.

Crowdfunding is an unbounding market thanks to the web context in which it is developed that allows entrepreneurs to enlarge their potential funding supply side. Our study offers relevant contribution especially for ethnic minority entrepreneurs and equity crowdfunding platforms to better understand potential investor behaviour and highlights the role of equity crowdfunding as tool for financial inclusion.

The paper is organised as follows. Section 2 reviews the relevant literature and presents the research questions. Section 3 presents data, variables and summary statistics used in the empirical section. Section 4 presents and discusses the results of the similarity effect on the amount invested by the single investors, with a study of their concurring role in its determination. Section 5 is the robustness section where we control if the seriality of the investor influence the result obtained. Finally, Section 6 concludes and discusses some implications and future development of this study.

## 2 "Similarity effect" in the investment process

For "similarity effect" we refer to the tendency of the members of a group to display a preference for associating with other members of the same group. This tendency is often described as "birds of a feather flock together". This pure preference-based mechanism highlights the role of personal similarity breeding connection and is a basic organizing principle presents across a wide range of social interaction contexts (McPherson et al., 2001). In financial literature, similarity effect is justified, among others, by arguments concerning: soft information, trust, and personal social network (Ruef et al., 2003; Hegde and Tumlinson, 2014; Bengtsson and Hsu, 2015; Hochberg et al. 2015). More generally in entrepreneurial finance, personal characteristics shared by seeker and investor reduces the cost of found and facilitates collaborations (Gompers et al., 2016; Alsos and Ljunggren, 2017). The influence of similarity biases is also important for understanding how venture capitalists decide whether to invest in specific companies. Several studies (Franke et al., 2006; Bruns et al., 2008; Murnieks et al., 2011) conclude that they tend to look more favourably at teams that have professional experience, human capital, ways of thinking similar to themselves.

Lazarsfeld and Merton (1954) distinguished two types of homophily: status homophily, in which similarity is based on informal, formal, or ascribed status, and value homophily, which is based on values, attitudes, and beliefs. Status homophily includes principal sociodemographic dimensions like race, ethnicity, sex, or age, and acquired characteristics like religion, education, occupation, or behaviour pattern.

Research in crowdfunding field focus especially on status homophily considering investors' gender. In reward-based projects and in peer-to-peer lending, women are more likely to

successful raise capital than male founders: female-led projects achieve a greater average pledge amount than male-led projects especially in sector whom they are historically underrepresented, as in the case for example of technological projects (Greenberg and Mollick, 2015). Female capital providers prefer to back projects founded by women due to choice homophily between female backers and entrepreneurs (Barasinska and Schäfer, 2014; Beier and Wagner, 2016; Marom et al., 2014) This result is confirmed also in case of equity crowdfunding campaigns. Vismara (2016c), analysing 58 equity offerings of UK crowdfunding platform Seedrs and comparing campaigns launched by firms with a female CEO with those launched by a male CEO, shows that on average, women invest 34 percent more than males. Therefore, the percentage of firms with a female CEO is higher in successful campaigns while the number of male investors is slightly higher in campaigns launched by male-led firms concluding that female investors strongly prefer firms led by females. Also Horvat and Papamarkou (2017) in their study on two UK equity crowdfunding platforms confirm on one hand that female-led campaigns enjoy higher success rate but on the other hand, they find that campaigns with higher participation rate from female investors tend to fail more at raising the target amount.

Within status homophily dimensions also ethnicity plays an important role. Members representing districts with significant immigrant and ethnic minority populations are more likely to support their minority colleagues emphasizing solidarity within ethnicities (Aldrich and Waldinger, 1990; Ruef et al., 2003). Ethnic solidarity theorists focus on social dynamics that facilitates the mobilization of ethnic resources for economic advancement in explaining why some immigrant-minority groups achieve economic success despite societal hostility and initial disadvantages. Thanks to ethnic solidarity, the economic resources support the enclave economy that is composed of businesses lead by immigrant groups which are concentrated in a distinct spatial location and that serve their own ethnic market and/or the general population (Portes and Manning, 1986; Portes and Jensen, 1989).

In VC selection process, a shared ethnicity between founder and VC increases the probability of an investment match and it has a high predictive power for early-stage investments (Hegde and Tumlinson, 2014; Bengtsson and Hsu, 2015) but it also influence the preference to collaborate with other venture capitalists (Gompers et al, 2016). Thanks to web context, crowdfunding can enlarge and facilitate the access to the pool of individuals with a similar ethnic characteristic. The similarity effect tend to get stronger as more types of relationships exist between two people, indicating that homophily on each type of relation cumulates to generate greater homophily (McPherson et al., 2001).

In the framework of the above literature review, we develop three principal research questions:

Rq1: To what extent gender and ethnic similarity affect the amount invested in the campaign?

*Rq2*: *Does the similarity effect change for different investors' gender and ethnic origin?* 

Rq3: Which is the relative importance of similarity effect on the variability of the amount invested in the campaign?

#### 3 Data and methods

#### 3.1 Sample

We collected data from Crowdcube, the largest crowdfunding platform in Britain (Beauhurst, 2017). Kaartemo, (2017) shows that that Crowdcube is relatively little researched in crowdfunding performance studies - mostly researchers have focused on Kickstarter that is a reward-based crowdfunding platform. Our sample includes the shareholder investors in 81 equity crowdfunding campaigns posted on Crowdcube from October 2011 to the end of October 2016. These campaigns were successfully concluded, closed, and all investments collected. We restrict our attention to these campaigns since these were the ones that enable us to find out the number of shareholders and the individual amount invested in a consistent manner.

We collected firm information about the campaign from the Crowdcube website. We were able to extract the name of investors and the number of shares subscribed by comparing information of the full list of shareholders from the model Annual Return, filled in accordance with Section 854 of the Companies Act 2006, at the date immediately before and immediately after the date of the campaign. The module AR101 is hosted in Companies House website. The price of the shares, when not disclosed directly in AR101 module, were obtained by dividing the amount collected during the equity crowdfunding campaign by the total number of shares subscribed by the investors during the campaign. The final sample includes 8,600 investments made by 5,996 unique personal investors. The dataset covers a timeframe of six years, from 2011 to 2016. Our sample is a broad cross-section of companies, industries and ages at the time of the equity crowdfunding campaigns (Table 1).

The average amount raised in our sample is £223,153, with a maximum of £1,9m and a minimum of £12,000. The average number of investors per campaign is equal to 106, the most crowded campaign involves 394 investors.

[INSERT TABLE 1 ABOUT HERE]

#### 3.2 Gender and ethnicity

#### Gender

Filled documents do not publish the gender of shareholders so to infer it we rely on automated methods. In our study we refer to *genderize.io tool* by comparing first names with a database including 86,710 distinct names across 81 languages (Greenberg and Mollick, 2015; Mohammadi and Shafi, 2017). The level of accuracy of this tool is 82%, the highest rate in the landscape of automated name-based gender detection methods widely used in scientific research (Karimi et al., 2016). Our final dataset includes 1,361 female users and 4,635 unique male investors. Also in the case of equity crowdfunding campaigns (Vismara, 2016c; Hervé et al.,2017) women investors are underrepresented; similar results hold also for venture and angel capital investments (Brush et al., 2004; Harrison and Mason, 2007).

In terms of gender of the entrepreneur, 18 out of 81 campaigns are led by at least one female proponent, and finally 63 are companies in which the single or all the proponents are male. Co-gender cases, cases for which the gender of backers and proponents matches, (Table 3) count for nearly the 68% of the recurrences. In the 1,946 investments made in female-companies, 21.8% of them (425) are associated to female investors. In the 6,654 investments made in male-companies, 82.3% of them (5,476) are associated to male investors.

The average amount invested by female investors is lower than amount provided by male investors: 1,766£ vs 2,179£, male investors invest on average 19% more than female investors. The same conclusion can be drawn by analysing the median amount invested (200£ vs 263£). These figures are opposite from the average bid reported by Vismara (2016c) from Seedrs where female investors invest on average 34% more than man.

#### **Ethnicity**

When ethnicity refers to a group, it implies that members have some awareness of group membership and a common national background and culture (Yinger, 1985). If ethnicity is linked to business founding, it represents a set of connections and social structure among members of an ethnic group that are attached to one another and the ways in which those social structures are used (Aldrich and Waldinger, 1990).

A number of different methodologies have been used to identify the ethnicity of the founder, from direct survey (Ruef et al., 2003; Wadwha et al., 2007; Hart and Acs, 2011), to post-secondary educations in their biographies (Kenney and Patton, 2015) to its inference from individual names (Hegde and Tumlinson, 2014; Bengtsson and Hsu, 2015). For identifying

investors' ethnicity, we refer to this last methodology looking for individual's surname on *forebears.io tool*, a genealogical sources, which includes information on surname geographic distribution. Regarding the entrepreneur, we use information conveyed by the platform and visible to the investors at the time of equity crowdfunding campaign. There are two principal drawbacks in our study. On the one hand, we have only the proponent information rather than all team at the time of the firm's founding, and thus the team question cannot be addressed directly. On the other hand, we cannot control for homonymy recurrences since we measure ethnicity coarsely, via surnames of individuals.

Individual investors' ethnicity is represented into 9 groups as follows<sup>2</sup>: African, Anglo-Saxon, Arabic, Asian, Oriental Bloc, European, Israelian, Southern Africa, Latin American. The same groups are used to identify proponents' ethnicity. Studies highlight that applicants from certain ethnic minority communities appear to experience greater problems than others in accessing external finance, bank loans at start-up and to have a higher propensity to turn to non-bank formal sources of start-up finance (Ram et al., 2002; Smallbone et al., 2003; Clark and Drinkwater, 2010). To understand the potential of availability of capital to traditionally underrepresented groups we focus our study on two main investor groups: the dominant group, investors that belong mainly to British and European origin ethnic groups (Anglo-Saxon, European, Oriental Block, Southern African ,Israelian) and the minority group, investors that belong to those that are considered underrepresented ethnic groups in the British and European financial market (Asian, African, Latin American, Arabic). The frequency for macro-ethnicity can be detached by analysing Table 2.

#### [INSERT TABLE 2 ABOUT HERE]

-

<sup>&</sup>lt;sup>2</sup> Africa (Angola, Benin, Burkina Faso, Cameroon, Congo, Ethiopia, Ghana, Ivory Coast, Kenya, Madagascar, Malawi, Mauritania, Mauritius, Mozambique, Nigeria, Senegal, Sierra Leone, Somalia, Sudan, Tanzania, Togo, Uganda, Zambia and Zimbabwe), Anglo-Saxon (Australia, Canada, England, Ireland, Malta, New Zealand and Unites States), Arabic (Algeria, Egypt, Iran, Iraq, Lebanon, Libya, Morocco, Oman, Pakistan, Saudi Arabia, Syria, Tunisia, Turkey and United Arab Emirates), Asian (Bangladesh, Cambodia, China, India, Indonesia, Japan, Malaysia, Mongolia, Myanmar, Nepal, North Korea, Philippines, Singapore, South Korea, Sri Lanka, Thailand and Vietnam), Eastern Bloc (Armenia, Azerbaijan, Bosnia and Herzegovina, Bulgaria, Croatia, Czech Republic, Estonia, Georgia, Hungary, Kazakhstan, Latvia, Lithuania, Moldova, Poland, Romania, Russia, Serbia, Slovakia, Slovenia, Tajikistan, Ukraine and Uzbekistan), Europe (Austria, Belgium, Cyprus, Denmark, Finland, France, Germany, Greece. Italy, Netherlands, Norway, Portugal, Spain, Sweden and Switzerland), Israel (Israel and Jordan), South Africa (South Africa and Swaziland), Latin America and Caribbean (Argentina, Barbados, Brazil, Chile, Colombia, Dominican Republic, Guatemala, Guyana, Haiti, Jamaica, Mexico, Paraguay, Peru, Trinidad and Tobago and Venezuela).

In terms of ethnicity, most investors are of Anglo-Saxon origins (69.8%), followed by Asian (9.4%) and European (8.1%) ones. African, Latin American, Oriental, Southern African and Israelian together are 10% of the sample. Co-ethnicity cases, cases for which the macroethnicity of backers and proponents are the same, (Table 3) count for nearly the 70% of the recurrences. Cases in which there is bot co-ethnicity and co-gender are 40% of the recurrences and where there is none of the two are 14% of the recurrences.

#### [INSERT TABLE 3 ABOUT HERE]

These results are consistent with Vismara (2016c), which bases his analysis on Seeders, another English platform, and shows that most investors are based in the UK (82.8%) but the distribution of investors in other countries is different from what one would expect in entrepreneurial finance, so from a geographical standpoint, equity crowdfunding allow company to deliver investment opportunities to a diversified set of investors.

### 3.3 Variable description

There are several observed characteristics that enable us to evaluate the investment likelihood by gender and ethnicity after having controlled for company, campaign, entrepreneur characteristics and investor characteristics.

In our analysis, we focus on the determinants of the amount invested by the crowd considering as dependent variables the logarithm of the amount invested by each investors in the equity crowdfunding campaign.

#### Independent variables

In the covariate group we consider three variables that explicit study the similarity between backers and proponents. In particular, gender similarity effect is proxied by the variable **Gender Similarity** which is a dummy variable equals 1 in case of gender similarity between backers and proponents. Ethnic similarity effect is proxied by the variable **Ethnic Similarity** which is a dummy variable equals 1 when backers and proponents share the same ethnicity. We also insert the interaction between ethnic and gender similarity - **Gender & Ethnic similarity** - is a dummy variable equals 1 in case of both gender and ethnic similarity between backers and proponents.

Variables related to the **entrepreneur** profile principally deal with their gender and ethnicity characteristics. **At least one female proponent** is a dummy variable equals 1 in case of at least one female proponents among the overall numbers of proponents and zero otherwise. The ethnic profile is captured by the variable **At least one ethnic minority proponent** is a dummy variable equals 1 in case of at least one ethnic minority proponent in the company and zero for dominant ones.

#### Control variables

There are several variables that might influence the investment decisions of investors, for which we have to control for. Some of them refer to company features, other to campaign structure and finally to investor characteristics.

Company features are insert in terms of firm age - the numbers of years from the firm's establishment to the date of the campaign. On average, the firm in the sample are 2.3 years old. The sector of activity captured by the variable Innovative – dummy variable equals one in case of companies active in highly innovative sectors isolated, as in Nesta (2009), on the basis of SIC code<sup>3</sup>; 18 companies (22.2%) in our sample belongs to highly innovative sectors. Finally, we insert the variable Team size computed as the log of number of people listed on the Crowdcube site to make up the group of executive around the entrepreneur. Smaller team may be riskier since they might be more recent and a higher number of board members is related to the number of investors and the funding amount reached during the campaign (Ahlers et al., 2015, Horvat and Papamarkou, 2017). In our sample the average team is formed by 3.2 people.

Campaign structure is proxied by the following variables. Equity offering the percentage of equity offered during the campaign. Equity offering can be considered as a proxy of firm risk for several reasons. On the one hand, signalling theory (Leland and Pyle, 1977) indicates the manager choice of raising equity as a negative signal for the investors since firms opportunistically choose to raise equity when managers know that their shares are overvalued.

\_

<sup>&</sup>lt;sup>3</sup>64.20; 72.20; 72.30; Telecommunication, Computer Programming and Software Service Industries; 72.40; 72.60; 73.10; Data Processing, Misc. Computer Services, R&D in Natural Sciences and Engineering; 30.01; 30.02; 32.20; 32.30; Office Equipment; Computers and other Information Processing Equipment; Television and Radio Transmitters and Apparatus for Line Telephony and Line Telegraphy; Television and Radio Receivers, Sound or Video Recording and Reproducing Apparatus; 33.20; 33.30; 33.40 Electronic Instruments and Appliances for Measuring, Checking (except Industrial Process Control); Electronic Industrial Process Control Equipment; Optical Instruments; Photographic Equipment; 24.41; 24.42; 33.10 Pharmaceutical Products and Preparations; Medical and Life Sciences Surgical Equipment and Orthopaedic Appliances; 24.16; 24.17; 31.10; Plastics and Synthetic Rubber in Primary Form; Electric Manufacturing; 31.20; 32.10; 35.30. Motors, Generators and Transformers; Electricity Distribution and Control Apparatus; Electronic Valves, Tubes and other Components; Aircraft and Spacecraft Manufacturing. (Nesta, 2009).

Likewise, an owner's decision to offer a lower amount of equity can indicate less adverse selection risk in that a bad outcome is less likely to be perceived by owners (Ahlers et al. 2015). Furthermore, greater equity offerings can dilute entrepreneurs' incentive to commit to their firms. Altogether, more equity offering suggests that a firm is riskier (Mohammadi and Shafi, 2017). On average, firms in our sample offers nearly 18% of equity; more or less the same (18.6%) in terms of investments. **Exit** is a dummy variable equals 1 if the exit strategy is explicit in the campaign and zero otherwise. A planned exit strategy influences the success of the campaign, receiving a return on the money it is typically only attained through an exit or liquidity event, so the presence of an exit strategy could make a campaign more attractive (Ahlers et al., 2015). In our sample, 65 campaigns (more than 80% of the total) envisage at least one exit option.

Finally, we control for the presence of serial investors in the sample and introducing the variable **Serial investor** that is a dummy variable equal to 1 for investors that have participated in more than one campaign and zero otherwise. The group of serial investors is 29% of the investors in the sample and the amount invested by serial investors is on average 794£.

#### [INSERT TABLE 4 ABOUT HERE]

#### 4. Similarity effect and investment size

We use Ordinary Least Square (OLS) regression to evaluate if similarity patterns, both in terms of gender and ethnicity, increases the individual investor investment size (Table 5). The dependent variable is the log of amount invested by the individual investor in a single campaign for the overall sample (Mod. 1). We also test whether similarity effect are etherogeneous by gender and ethnic groups of the investors. We show estimates for the subsample of female/male investors (Mod. 2/3), for the sub-sample of Dominant/Minority ethnic group investors (Mod. 4/5). Finally for the combination of gender and ethnic group of the investors: female investors that come from a dominant/minority ethnic group (Mod. 8/9).

Amount invested = 
$$\alpha + \beta_1 X + \beta_2 Controls + \varepsilon$$
 [2]

[INSERT TABLE 5 ABOUT HERE]

For the full sample, estimation results show that gender and ethnic similarity individually have no significant effect on the amount invested, whereas we find that the coexistence of gender and ethnic similarity determines a 30% increase of the amount invested. Moreover, results highlight that the amount invested decreases of 40% when the campaign has at least one female proponent and conversely, the amount invested raises 41% when the campaign has a proponent belonging at one ethnic minority group.

The amount invested in equity crowdfunding campaigns (Model 1) is higher in older company with a larger number of members in the team. Concerning the variables related to the campaigns, those that receive a larger invested amount state a presence of an exit strategy. The percentage of equity offered in the campaign is not significant. Thus, the level of amount invested do not change by campaign and company characteristics but there are differences when we consider investors' and entrepreneurs' characteristics.

Serial investors confirm descriptive statistics that show a lower amount invested and high frequency of the investment regardless gender and ethnic origin of the investors considered. This investment behavior seem to be similar to a gambling strategy.

For female investors (Model 2), gender similarity negatively affects the amount invested otherwise for male investors there is a positive effect (Model 3). In other words, it seems that male investors look at gender similarity is in incentive to invest a higher amount of money while the opposite holds for female investors. This result appears to be consistent with literature that suggest that women in male-dominated fields prefer to support members of the dominant group (Greenberg and Mollick, 2015). Also in the business angel market, research has demonstrated the tendency for women angels to refuse financing to women entrepreneurs, although they are more likely to seek funding from other women (Becker-Blease and Sohl 2007). In case of male investors the interaction fosters the amount invested. Focusing to the comparison between female and male investors, it seems that there are no straight differences in the types of companies and campaigns characteristics that condition the size of investments (Models 2 and 3). The size of their investments increases in older companies, with larger team and when an exit strategy is considered.

Similarly to the comparison by gender, results show different effects when we distinguish investors belonging to different ethnic groups (Models 4 and 5) and when we match both gender and ethnic group of investors (Models 6, 7, 8 and 9). In particular, ethjnic similarity have a negative effect on the amount invested by investors belonging to a dominant ethnic group while they have a positive effect on ethnic minorities. For investors that belong to an ethnic minority (Model 5) the presence of an ethnic similarity with the proponent seems to

compensate other aspects of the campaign, for example: the team dimension and the absence of an exit strategy.

Comparing female investors by their ethnicity group (Model 6 and 7) we find that both of them tend to invest significantly less when there is a female proponent but the overall similarity effect is over turned for female investors belonging to an ethnic minority group. In models 8 and 9, male investors that belong to the dominant ethnic group there is a negative and significant effect on the amount invested in company run by an entrepreneur of the same origin. For male minority, ethnic and gender similarity have a strong and positive effect on the amount invested but the overall similarity effect is reduced when this similarity appears together. This can be linked to the fact that for male minority investors the amount invested is lower in companies that are active in innovative sector and the most part of ethnic minority companies are innovative businesses.

Results showed by heterogeneous effects indicate different trends for similarity effect and they depend on investors' characteristics.

About Rq1, it seems that similarity effect influence the amount invested by the individual investor and its result change according to investors gender and ethnic origin (Rq2). Nonetheless, the different covariates introduced, being heterogeneous, as confirmed by the variables' descriptions, do not share a common unit of measure. This implies that their relative contribution can be observed uniquely in terms of their relevance, in a multivariate frame, to explaining a given object of analysis.

To better point out their concurring role in the definition of the investment size, we investigate the importance and the relative contribution of different aspects that characterized an equity crowdfunding investment, for explaining the quantity of amount invested by the single investor. The characteristics identified can be associated to company, entrepreneur, campaign, investor and degree of similarity in terms of gender and ethnicity. Coherently, we model in equation (2) the relationship between the identified characteristics and the amount invested (in log).

In the general equation (2) we follow a Shapley decomposition method (Shapley, 1953) that, since its introduction in game theory, permits distributing a goodness-of-fit measure of an econometric model among regressors. As a goodness-of-fit measure, we adopt the total R2-value obtained from the OLS pooled estimations of equation (2). Since some attributes are formed by groups of variables, we use a generalization of the Shapley value, the Owen value (Owen, 1977), which allows decomposition in the case of exogenously grouped regressors, as

suggested by Shorrocks (2013). Results of the Shapley and Owen value decomposition are reported in Table 6.

#### [INSERT TABLE 6 ABOUT HERE]

As expected, being a serial investor has an important role in explaining the existing heterogeneity in the amount invested in the campaign and this is particular true for male investors both belonging to dominant or minority ethnic group (Table 6).

Results underline the pivotal role that company characteristics have on the amount invested, a role nearly exclusively related to company's age and structure of the team. Turning to the remaining characteristics, the overall R2 of the full sample can be explained mainly by two groups of characteristics: entrepreneur profile, linked to the presence of at least one female entrepreneur in the company, and the combination of gender and ethnic similarities between backers and proponents that condition the variability of the amount invested in an unneglectable manner (Rq3). Conversely, campaign features appears to be the weakest characteristics in terms of explanatory power with respect to the size of the investment.

As in the previous analysis, we disaggregate our sample by investor gender and ethnic origin (Table 7). Female and male investors look at different aspects in the campaign. Even if company characteristics still remain the most important aspect that influence the variability of the amount invested, female investors give more importance to the size of the team whereas male investors tend to increase the investment size in older companies. The similarity group represents a main aspect for all investor categories in particular for female investors for which in case of minority ethnic profile contributes for nearly 22% to the formation of the overall R2. For male that belong to an ethnic minority, gender similarity assumes the same importance of company age. The seriality profile strongly conditions the variability of the amount invested for male investors belonging to dominant ethnic group. This characteristic explains more the 55% of the variance of the amount invested.

#### [INSERT TABLE 7 ABOUT HERE]

## 5. Robustness

Given the importance of the serial investors, we decide to verify if the investment behavior adopted by this kind of investors influence the overall previous results.

Table 8 and Table 9 present the results of our estimations without considering serial investors in the sample. Our mains findings remain qualitatively unchanged and the stability of the coefficients in the model estimation confirm the robustness of our results.

#### [INSERT TABLE 8 AND 9 ABOUT HERE]

#### 6. Conclusion

This study explores the presence of ethnic and gender similarities between backers and proponents in equity crowdfunding investment investigating the relationship between similarity patterns and the size of investment. Our results show that similarity, in terms of both gender and ethnic matching between investors and proponent, is an important aspect in the investment decision but its effects change for the different investors' characteristics. In particular, differently from reward based crowdfunding, female investors are less present in equity crowdfunding market and the size of female investment is lower than those of male ones. Female investors show a clear preference for male lead campaigns whereas male investors tend to support more male company; nevertheless in case of female entrepreneur belong to an ethnic minority receive the support of female investors sharing the same ethnic origin of the proponent.

We find that equity crowdfunding appears to reduce constraints for founders that belong to an ethnic minority seeking capital, both for female and male investors the level of engagement in the campaign is higher if they share the same ethnic minority profile of the proponent. An ethnic match especially for ethnic minority group compensates other campaigns' aspects, for example the number of people in the company's team and the presence of an exit strategy. The combination of gender and ethnic similarities between backers and proponents conditions the variability of the amount invested in an unneglectable manner whereas campaigns' characteristics appear to be not relevant.

There are several advantages of this setting for studying the impact of gender and ethnic similarities in investment decisions. First, unlike much of the past work on this topic that mainly rests on gender profile, our results regarding ethnic similarity represents a novel in this field. Second, studying similarity effect in crowdfunding setting, contributes to a better understanding of the persistence of this phenomenon also in the web context and in online financing relationships.

Finally, we contribute to the nascent literature in equity crowdfunding and financial inclusion related fields. Financial inclusion in fact, is defined as a state in which individuals or firms

have effective access to financial products and services appropriate to their needs (Demirgüç-Kunt et al., 2015). Equity crowdfunding appears to be a tool for a sustainable financial inclusion of ethnic minorities in the financial system, both from seekers and suppliers of capital. From seekers side, our study allows women and ethnic minority entrepreneurs to identify investors' behaviours and characteristics that maximize fund raising opportunities through equity crowdfunding campaigns. From suppliers side we provide a snapshot of how women and ethnic minority perceive and invest in equity crowdfunding projects adopting a longitudinal perspective and contributing to theory building. For equity crowdfunding literature we show that the Investment decision in equity crowdfunding are driven not only by financial motivations but also by a similarity matching between proponent and investors. And this is particular true for female investors belonging to an ethnic minority group.

From a practical perspective, the results are relevant not only for female and ethnic minority but for all entrepreneurs, investors, and crowdfunding platforms alike, as understanding crowd composition and investment dynamics are in the interest of each group. Such knowledge may have a positive impact on the probability of campaign success, which especially for equity model, studies show different results (Lukkarinen et al, 2016).

Due to a different result compared with reward based platform (Greenberg and Mollick, 2015) future researches could extend analysis to other platforms or compare different crowdfunding model (i.e.: reward based vs equity based) to understand: why in riskier situation female investors prefer male business? Or, do female investors prefer to support female companies only in a low risk context? In particular, about the emotional part of equity crowdfunding campaign, female and male investors may have different investment behaviour with words and language used to present the project. Do ethnic minority and female entrepreneurs adopt specific words for promoting the crowdfunding campaign? We hope that future works could go in- depth to these issues.

## **Bibliography**

- Ahlers, G. K. C., Cumming, D., Günther, C., & Schweizer, D. (2015). Signaling in Equity Crowdfunding. *Entrepreneurship: Theory and Practice*, *39*(4), 955–980. https://doi.org/10.1111/etap.12157
- Aldrich, H. E., & Waldinger, R. (1990). Ethnicity and Entrepreneurship. *Annual Review of Sociology*, *16*(1), 111–135. https://doi.org/10.1146/annurev.soc.16.1.111
- Alsos, G. A., & Ljunggren, E. (2017). The Role of Gender in Entrepreneur–Investor Relationships: A Signaling Theory Approach. *Entrepreneurship: Theory and Practice*, 41(4), 567–590. https://doi.org/10.1111/etap.12226
- Barasinska, N., & Schäfer, D. (2014). Is Crowdfunding Different? Evidence on the Relation between Gender and Funding Success from a German Peer-to-Peer Lending Platform. *German Economic Review*, *15*(4), 436–452. https://doi.org/10.1111/geer.12052
- Beauhurst, Crowdfunding Index, Crowdfunding in the UK Q1 2017. https://about.beauhurst.com/research/the-deal/
- Becker-Blease, J. R., & Sohl, J. E. (2007). Do women-owned businesses have equal access to angel capital? Journal of Business Venturing, 22(4), 503–521. https://doi.org/10.1016/j.jbusvent.2006.06.003
- Bengtsson, O., & Hsu, D. H. (2015). Ethnic matching in the U.S. venture capital market. Journal of Business Venturing, 30(2), 338–354. https://doi.org/10.1016/j.jbusvent.2014.09.001
- Beier, M. & K. Wagner (2016), 'User behavior in crowdfunding platforms-Exploratory evidence from Switzerland', 49th Hawaii International Conference on System Sciences accessed 12, December 2017 at
- http://ieeexplore.ieee.org/stamp/stamp.jsp?arnumber=7427632&tag=1
- Bruns, V., Holland, D. V., Shepherd, D. A., & Wiklund, J. (2008). The role of human capital in loan officers' decision policies. *Entrepreneurship: Theory and Practice*, *32*(3), 485–506. https://doi.org/10.1111/j.1540-6520.2008.00237.x

- Brush, C. G., Carter, N. M., Gatewood, E. J., Greene, P. G., & Hart, M. (2004). Gatekeepers of Venture Growth: A Diana Project Report on the Role and Participation of Women in the Venture Capital Industry.

  \*\*SSRN Electronic Journal.\*\* https://doi.org/10.2139/ssrn.1260385
- Cholakova, M., & Clarysse, B. (2015). Does the Possibility to Make Equity Investments in Crowdfunding Projects Crowd Out Reward-Based Investments? *Entrepreneurship Theory and Practice*, 39(1), 145–172. <a href="https://doi.org/10.1111/etap.12139">https://doi.org/10.1111/etap.12139</a>
- Clark, K., & Drinkwater, S. (2010). Recent trends in minority ethnic entrepreneurship in Britain. International Small Business Journal. https://doi.org/10.1177/0266242609355831
- Collins, L., & Pierrakis, Y. (2012). The venture crowd: Crowdfunding equity investment into business. *Nesta*, (July), 1–36. Retrieved from http://scholar.google.com/scholar?hl=en&btnG=Search&q=intitle:The+venture+crowd+crowdfunding+equity+investment+into+business#4
- Demirgüç-Kunt, A., Klapper, L. F., Singer, D., & Van Oudheusden, P. (2015). The global findex database 2014: Measuring financial inclusion around the world. Available at: https://openknowledge.worldbank.org/bitstream/handle/10986/21865/WPS7255.pdf?seq uence=2
- Franke, N., Gruber, M., Harhoff, D., & Henkel, J. (2006). What you are is what you like-similarity biases in venture capitalists' evaluations of start-up teams. *Journal of Business Venturing*, *21*(6), 802–826. https://doi.org/10.1016/j.jbusvent.2005.07.001
- Gerber, E. M., Hui, J. S., & Kuo, P.-Y. (2012). Crowdfunding: Why people are motivated to post and fund projects on crowdfunding platforms. In *Proceedings of the International Workshop on Design, Influence, and Social Technologies: Techniques, Impacts and Ethics 2(11)*. https://doi.org/http://dx.doi.org/10.1145/2530540
- Gompers, P. A., Mukharlyamov, V., & Xuan, Y. (2016). The cost of friendship. *Journal of Financial Economics*, 119(3), 626–644. https://doi.org/10.1016/j.jfineco.2016.01.013

- Greenberg, J., & Mollick, E. (2015). Leaning in or leaning on? Gender, Homophily, and activism in crowdfunding. Academy of Management Proceedings. https://doi.org/10.5465/AMBPP.2015.18365abstract
- Harrison, R. T., & Mason, C. M. (2007). Does gender matter? Women business angels and the supply of entrepreneurial finance. *Entrepreneurship: Theory and Practice*, 31(3), 445–472. https://doi.org/10.1111/j.1540-6520.2007.00182.x
- Hart, D. M., & Acs, Z. J. (2011). High-tech immigrant entrepreneurship in the united states. *Economic Development Quarterly*, 25(2), 116–129. https://doi.org/10.1177/0891242410394336
- Hegde, D., & Tumlinson, J. (2014). Does Social Proximity Enhance Business Partnerships? Theory and Evidence from Ethnicity's Role in U.S. Venture Capital. *Management Science*, 60(9), 2355–2380. https://doi.org/10.1287/mnsc.2013.1878
- Hervé, F., Manthé, E., Sannajust, A., & Schwienbacher, A. (2017), Determinants of Individual Investment Decisions in Investment-Based Crowdfunding. Available at SSRN: https://ssrn.com/abstract=2746398 or http://dx.doi.org/10.2139/ssrn.2746398.
- Hochberg, Y. V., Lindsey, L. A., & Westerfield, M. M. (2015). Resource accumulation through economic ties: Evidence from venture capital. *Journal of Financial Economics*, 118(2), 245–267. https://doi.org/10.1016/j.jfineco.2015.06.008
- Horvat, E. A. & Papamarkou, T. (2017) Gender Differences in Equity Crowdfunding. In: *HCOMP 2017*, Quebec City, QC, Canada, 24-26 Oct 2017.
- Karimi, F., Wagner, C., Lemmerich, F., Jadidi, M., & Strohmaier, M. (2016, April). Inferring gender from names on the web: A comparative evaluation of gender detection methods. In Proceedings of the 25th International Conference Companion on World Wide Web (pp. 53-54). International World Wide Web Conferences Steering Committee
- Kaartemo, V. (2017). The elements of a successful crowdfunding campaign: A systematic literature review of crowdfunding performance. International Review of Entrepreneurship.

- Kenney, M., & Patton, D. (2015). Gender, ethnicity and entrepreneurship in initial public offerings: Illustrations from an open database\*. *Research Policy*, 44(9), 1773–1784. https://doi.org/10.1016/j.respol.2015.01.003
- Lazarsfeld, P. F., & Merton, R. K. (1954). Friendship as a Social Process: A Substantive and Methodological analysis. *Freedom and Control in Modern Society*, *18*, 18–66. https://doi.org/10.1111/j.1467-8705.2012.02056\_3.x
- Leitch, C., Welter, F., & Henry, C. (2018). Women entrepreneurs' financing revisited: taking stock and looking forward: New perspectives on women entrepreneurs and finance (Special Issue). *Venture Capital*, 20(2), 103–114. https://doi.org/10.1080/13691066.2018.1418624
- Leland, H. E., & Pyle, D. H. (1977). Informational asymmetries, financial structure, and financial intermediation. *Journal of Finance*. https://doi.org/10.2307/2329297
- Lukkarinen, A., Teich, J. E., Wallenius, H., & Wallenius, J. (2016). Success drivers of online equity crowdfunding campaigns. *Decision Support Systems*, 87, 26–38. https://doi.org/10.1016/j.dss.2016.04.006
- Marom, D., Robb, A., & Sade, O. (2014). Gender dynamics in crowdfunding (Kickstarter). Evidence on Entrepreneurs, Investors, Deals and Taste-Based Discrimination. Available at SSRN: https://ssrn.com/abstract=2442954 or http://dx.doi.org/10.2139/ssrn.2442954
- McPherson, M., Smith-Lovin, L., & Cook, J. M. (2001). Birds of a Feather: Homophily in Social Networks. *Annual Review of Sociology*, 27(1), 415–444. https://doi.org/10.1146/annurev.soc.27.1.415
- Mohammadi, A., & Shafi, K. (2017). Gender differences in the contribution patterns of equity-crowdfunding investors. Small Business Economics, 50(2), 275–287. https://doi.org/10.1007/s11187-016-9825-7
- Murnieks, C. Y., Haynie, J. M., Wiltbank, R. E., & Harting, T. (2011). "I like how you think": Similarity as an interaction bias in the investor-entrepreneur dyad. *Journal of Management Studies*, 48(7), 1533–1561. https://doi.org/10.1111/j.1467-6486.2010.00992.x

- Ordanini, A., Miceli, L., Pizzetti, M., & Parasuraman, A. (2011). Crowd-funding: transforming customers into investors through innovative service platforms. *Journal of Service Management*, 22(4), 443–470. https://doi.org/10.1108/09564231111155079
- Owen, G. (1977). Values of games with a priori unions. In R. Henn, O. Moeschlin (eds.), Mathematical Economics and Game Theory. Lecture Notes in Economics and Mathematical Systems, 76–88. Berlin: Springer.
- Polzin, F., Sanders, M., & Stavlöt, U. (2018). Do investors and entrepreneurs match? Evidence from The Netherlands and Sweden. *Technological Forecasting and Social Change*, 127, 112–126. https://doi.org/10.1016/j.techfore.2017.07.016
- Portes, A., & Manning, R. (1986). The Immigrant Enclave: Theory and Empirical Examples. Competitive Ethnic Relations.
- Portes, A., & Jensen, L. (1989). The Enclave and the Entrants: Patterns of Ethnic Enterprise in Miami before and after Mariel. American Sociological Review. https://doi.org/10.2307/2096244
- Ram M, Smallbone D. and Deakins D. (2002). The Finance and Business Support Needs of Ethnic Minority Firms in Britain. London: British Bankers Association
- Roper, S., Hales, C., Bryson, J. R., & Love, J. (2009). Measuring sectoral innovation capability in nine areas of the UK economy. NESTA, London.
- Ruef, M., Aldrich, H. E., & Carter, N. M. (2003). The Structure of Founding Teams: Homophily, Strong Ties, and Isolation among U.S. Entrepreneurs. American Sociological Review, 68(2), 195. https://doi.org/10.2307/1519766
- Schwienbacher, A., & Larralde, B. (2010). Crowdfunding of Small Entrepreneurial Ventures. SSRN Electronic Journal. https://doi.org/10.2139/ssrn.1699183
- Shapley, L. S. (1953). A value for n-person games. In H. Kuhn, A. Tucker (eds.), *Contributions to the Theory of Games* (AM-28), Volume II, 307-317. Princeton, NJ: Princeton University Press.

- Shorrocks, A. F. (2013). Decomposition procedures for distributional analysis: a unified framework based on the Shapley value. *Journal of Economic Inequality*, 11(1), 99-126.
- Smallbone, D., Ram, M., Deakins, D., & Baldock, R. (2003). Access to Finance by Ethnic Minority Businesses in the UK. International Small Business Journal. https://doi.org/10.1177/02662426030213003
- Vismara, S. (2016a). Information Cascades Among Investors in Equity Crowdfunding. *Entrepreneurship: Theory and Practice*. https://doi.org/10.1111/etap.12261
- Vismara, S. (2016b). Equity retention and social network theory in equity crowdfunding. Small Business Economics, 46(4), 579–590. https://doi.org/10.1007/s11187-016-9710-4
- Vismara, S. (2016c). Gender in Entrepreneurial Finance: Matching Investors and Entrepreneurs in Equity Crowdfunding, Forthcoming in Albert Link, eds., "Gender and Entrepreneurial Activity", Cheltenham, UK: Edward Elgar
- Yinger, J. M. (1985). Ethnicity. *Annual Review of Sociology*, 11, 151–180. https://doi.org/10.2307/2083290

**Table 1 - Descriptive statistics** 

Panel A: Sample overview	Obs					
Number of companies (ECF campaigns)	81					
Number of investments made by personal investors	8,600					
Number of investments made by female investors	1,603					
Number of investments made by male investors	6,997					
Number of unique female investors	1,361					
Number of unique male investors	4,635					
Number of investments made by ethnic minority investors	1,568					
Number of investments made by ethnic dominant investors	7,032					
Number of investments made by a serial investors	2,526					
Number of unique ethnic minority investors	1,064					
Number of unique ethnic dominant investors	4,932					
Panel B: Summary statistics of key variables for companies	Obs	mean	min	median	max	sd
Company age at the date of ECF (in years)	81	2.3	0.0	2.0	11.0	2.3
Company sector Innovative (1=yes; 0= no)	81	0.22				0.42
Team size (number)	81	3.20	0.0	3	12	3.18
Panel C: Summary statistics of key variables for campaigns	Obs	mean	min	median	max	sd
Total amount raised (£)	81	223,153	12,000	139,900	1,962,730	223,153
Total equity offer (in %)	81	17.7%	4.0%	16.8%	48.0%	0.08
Exit strategies (1=yes; 0= no)	81	0.80	0	1.00	1.00	0.40
Panel D: Summary statistics of key variables for entrepreneur	Obs	mean	min	median	max	sd
Proponents (number)	81	1.3	1.0	1.0	3.0	0.5
At least 1 female proponents	81	0.22	0	0	1.0	0.42
At least 1 ethnic minority proponent	81	0.19	0	0	1.0	0.39
Panel E: Summary statistics of key variables for investors	Obs	mean	min	median	max	Sd
Investors per campaign (number)	81	106	8	83	394	81
Female investors per campaign (number)	81	20	1	12	125	24
Fraction of female investors per campaign (%) Fraction of dollar amount female investors invested per campaign	81	17.5%	3.6%	15.4%	45.6%	0.09
(%)	81	23.0%	1.7%	17.3%	50.6%	0.17
Ethnic minority investors per campaign (number)	81	19	1	15	75	16.73
Fraction of ethnic minority investors per campaign (%) Fraction of dollar amount ethnic minority investors invested per	81	17.5%	6.3%	17.4%	35.5%	0.06
campaign (%)	81	19.9%	0.9%	18.8%	70.7%	0.16
Dollar amount invested by personal investors (£)	8,600	2,102	0.23	250	1,394,612	16,186
Dollar amount invested by female investors (£)	1,603	1,766	2.3	200	139,461	6,284
Dollar amount invested by male investors (£)	6,997	2,179	0.23	263	1,394,612	17,690
Dollar amount invested by ethnic minority investors $(\mathfrak{L})$	2,666	1,724	5.00	200	96,666	5,499
Dollar amount invested by dominant investors (£)	5,923	2,163	0.23	263	1,394,612	17,709
Dollar amount invested by serial investors (£)	2,526	794	0.23	147	51,560	2,713

Table 2 - Frequency of ethnicity groups in the sample

Ethnicity	Inv	restors	Pro	ponents
Ethnicity	Obs	%	Obs	%
Dominant				
Anglo-Saxon	4.187	69,8%	77	73,3%
European	484	8,1%	7	6,7%
Oriental Block	176	2,9%	1	1,0%
Southern African	63	1,1%	1	1,0%
Israelian	22	0,4%	3	2,9%
Minority				
Asian	562	9,4%	11	10,5%
African	189	3,2%	2	1,9%
Latin American	157	2,6%	1	1,0%
Arabic	156	2,6%	2	1,9%
Total	5.996	100,0%	105	

Table 3 - Gender similarity and ethnic similarity

Panel A: Gender Similarity			
T. 1	Female p	TC 4.1	
Female investor	No	Yes	Total
	5.476	1.521	6.997
No	82,3%	78,2%	81,4%
Yes	1.178	425	1.603
ies	17,7%	21,8%	18,6%
Total	6.654	1.946	8.600
Total	100,0%	100,0%	100,0%
Panel B: Ethnic Similarity			
T	Proponent	t ethnicity	Total
Investor ethnicity	Dominant	Minority	Total
	5.684	1.348	7.032
Dominant	82,4%	78,7%	81,8%
Minority -	1.203	365	1.568
Willionty  -	17,6%	21,3%	18,2%
Total	6.887	1.713	8.600
Total	100,0%	100,0%	100,0%
Panel C: Gender and Ethnic Similarities			
0 1 1 1 1	Ethnic si	milarity	Tr. 4.1
Gender similarity	No	Yes	Total
	1.254	2.414	3.668
No _	46,5%	40,9%	42,7%
Yes	1.445	3.487	4.932
168	53,5%	59,1%	57,4%
Total	2.699	5.901	8.600
Total	100,0%	100,0%	100,0%

**Table 4 – Correlation matrix** 

Variables	Log amount inv.	Gender similarity	Ethnic similarity	At least one female proponent	At least one ethnic minority proponent	Innovative	Firm Age	Team Size	Equity offering	Exit	Serial Investor
Ln(Amount invested)	1										
Gender similarity	0.0711*	1									
Ethnic similarity	0.0507*	0.0521*	1								
At least one female prop.	-0.1072*	-0.5451*	-0.1146*	1							
At least one ethnic minority proponent	-0,0175	-0.0774*	-0.3634*	0.0664*	1						
Innovative	-0.0316*	0.0261*	0,0074	-0.0941*	-0.0269*	1					
Firm age	0.1310*	0.0275*	0.0327*	-0.0339*	-0.1944*	0,0052	1				
Team size	-0.0959*	-0.0865*	0,004	0.0845*	0.1346*	0.0946*	-0.3396*	1			
Equity offering	-0,0177	0,0068	-0.0976*	-0.0233*	0.2650*	-0.1909*	-0.2601*	-0,0202	1		
Exit	-0.0286*	-0.0434*	-0.0265*	0.1405*	0.1124*	-0,0179	-0.0542*	0.5604*	0.0787*	1	
Serial investor	-0.1778*	0.0213*	-0.0545*	0.0405*	0.0427*	0.0402*	-0.0667*	-0,0196	0.0490*	0.0545*	1

 $Table \ 5-Similarity \ effect \ by \ gender \ and \ ethnicity \ groups \ of \ investors$ 

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Variables	All sample	Female	Male	Dominant	Minority	Female - Dominant	Female - Minority	Male - Dominant	Male - Minority
Gender similarity	-0.089	-0.677***	0.236***	-0.296***	0.288*	-0.752***	-0.931***	0.008	0.737***
	(0.079)	(0.180)	(0.083)	(0.095)	(0.151)	(0.213)	(0.337)	(0.101)	(0.156)
Ethnic similarity	-0.049	0.116	-0.057	-0.435***	0.696***	-0.105	0.136	-0.552***	1.170***
	(0.082)	(0.136)	(0.108)	(0.093)	(0.206)	(0.148)	(0.379)	(0.125)	(0.253)
Interaction [gender & ethnic	0.285***	0.114	0.284**	0.510***	-0.259	-0.027	2.197***	0.645***	-0.976***
similarity]	(0.095)	(0.239)	(0.118)	(0.111)	(0.249)	(0.270)	(0.712)	(0.140)	(0.297)
At least one female	-0.409***			-0.479***	-0.299**				
proponent	(0.065)			(0.073)	(0.151)				
At least one ethnic minority	0.412***	0.488***	0.377***						
proponent	(0.062)	(0.156)	(0.067)						
Innovative	-0.088	-0.031	-0.089	-0.039	-0.374**	-0.005	-0.298	-0.037	-0.368**
	(0.063)	(0.158)	(0.069)	(0.069)	(0.150)	(0.175)	(0.368)	(0.075)	(0.160)
Firm age	0.122***	0.073***	0.132***	0.121***	0.104***	0.083***	-0.010	0.130***	0.130***
	(0.010)	(0.023)	(0.011)	(0.011)	(0.024)	(0.024)	(0.058)	(0.012)	(0.026)
Team size	0.245***	0.191*	0.273***	0.288***	0.130	0.266**	-0.120	0.333***	0.281**
	(0.042)	(0.114)	(0.047)	(0.047)	(0.106)	(0.121)	(0.321)	(0.054)	(0.114)
Equity offering	0.058	0.473	0.040	0.390	-0.360	1.148	-0.638	0.347	-0.291
	(0.273)	(0.767)	(0.294)	(0.296)	(0.657)	(0.860)	(1.747)	(0.319)	(0.707)
Exit	0.697***	0.838***	0.686***	0.795***	0.070	0.999***	0.035	0.776***	0.179
	(0.134)	(0.293)	(0.150)	(0.140)	(0.426)	(0.289)	(0.861)	(0.158)	(0.485)
Serial investor	-0.799***	-0.739***	-0.862***	-0.748***	-0.967***	-0.580**	-1.250***	-0.837***	-0.952***
	(0.086)	(0.219)	(0.093)	(0.101)	(0.151)	(0.255)	(0.418)	(0.109)	(0.160)
Constant	5.613***	5.528***	5.289***	5.857***	5.622***	5.524***	6.142***	5.537***	5.013***
	(0.117)	(0.255)	(0.128)	(0.135)	(0.248)	(0.308)	(0.546)	(0.146)	(0.278)
Year fixed effect	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	8.600	1.603	6.997	7.032	1.568	1.336	267	5.696	1.301
R-squared	0,098	0,121	0,100	0,089	0,140	0,115	0,224	0,092	0,144
Adj. R-squared	0,096	0,114	0,098	0,087	0,132	0,107	0,188	0,090	0,136

Notes: This table reports the results of a OLS regression. The dependent variable is the log of amount invested by the individual investor in a single campaign for the overall sample (Mod. 1), for the sub-sample of female investors (Mod. 2) and male investors (Mod. 3), for the sub-sample of dominant investors (Mod. 4), for minority investors (Mod. 5) and the combination of gender and ethinicity of the investors: female investors that belong to the dominant ethnicity (Mod. 6), female investors that belong to the minority ethnicity (Mod. 7); male investors that belong to the dominant ethnicity (Mod. 8) and male investors that belong to the minority ethnicity (Mod. 9). The following company features' controls are included in the regression: Firm age is the the numbers of years from the firm's establishment to the date of the campaign, Innovative is a dummy variable equals one in case of companies active in highly innovative sectors, Team size computed as the log of number of people listed on the site to make up the group of executive around the entrepreneur. Two campaign structure controls are included as follows: Equity offering is the percentage of equity offered during the campaign, Exit is a dummy variable equals 1 if the exit strategy is explicit in the campaign and zero otherwise. The control variables related to the entrepreneur characteristics are the following: At least one female proponent is a dummy variable equals 1 in case of at least one female proponents among the overall numbers of proponents and zero otherwise. Variables related to the investor profile is Serial investor is a dummy variable equals 1 in case of at least one ethnic minority proponents of proponents and zero otherwise. Similarity effect - Gender similarity - is a dummy variable equals 1 in case of gender similarity between backers and proponents, ethnical similarity effect - Ethnic similarity - is a dummy variable equals 1 in case of gender similarity - Gender similarity & Ethnic similarity - is a dummy variable equals 1 in case of g

Table 6 – Shapley and Owen value decomposition of R2

Cassa	Variable	Owen value				
Group	variable	Ind. %R2	Group %R2			
	Gender similarity	2,34	8,13			
	Ethnic similarity	1,50				
Similarity	Gender & Ethnic similarity	4,29				
Entrepreneuer	At least one female proponent	12,68	13,41			
characteristics	At least one ethnic minority proponent	0,74				
C	Innovative	1,44	30,06			
Company characteristics	Firm age	19,03				
characteristics	Team size	9,59				
Campaign	Equity offering	0,19	1,12			
characteristics	Exit	0,93				
Investor characteristics	Serial investor	47,27	47,27			

Note: Under the general equation of (2), this table reports the fraction of the overall R-squared (per cent), explained by the different groups of characteristics — Owen value (Gr) — and the individual share (Ind) of the overall R-squared (per cent) associated with each variable belonging to a specific characteristic (Shapley decomposition). Note that these figures are median values within bootstrap percentile confidence intervals (level of confidence: 90%). This explains cases where Total is not always 100.00.

The following company characteristics are included in the regression: Firm age is the the numbers of years from the firm's establishment to the date of the campaign, Innovative is a dummy variable equals one in case of companies active in highly innovative sectors, Team size computed as the log of number of people listed on the site to make up the group of executive around the entrepreneur. Two campaign structure characteristics are included as follows: Equity offering is the percentage of equity offered during the campaign, **Exit** is a dummy variable equals 1 if the exit strategy is explicit in the campaign and zero otherwise. The variables related to the entrepreneur characteristics are the following: At least one female proponent is a dummy variable equals 1 in case of at least one female proponents among the overall numbers of proponents and zero otherwise, At least one ethnic minority proponent is a dummy variable equals 1 in case of at least one ethnic minority proponents among the overall numbers of proponents and zero otherwise. Variables related to the investor characteristics is Serial investor is a dummy variable equal to 1 for serial investors and zero for otherwise. Similarity effects rest on three variables; gender similarity effect - Gender similarity - is a dummy variable equals 1 in case of gender similarity between backers and proponents, ethnical similarity effect - Ethnic similarity - is a dummy variable equals 1 when backers and proponents share the same ethnicity and the interaction between ethnic and gender similarity - Gender similarity & Ethnic similarity - is a dummy variable equals 1 in case of gender and ethnic similarity between backers and proponents. Year fixed effects are considered in the model.

 $Table \ 7-Shapley \ and \ Owen \ value \ decomposition \ of \ R2 \ by \ investors \ characteristics$ 

		Owen value									
Group	- Variable	Fema	ale	Ma	ale	Do	minant	M	inority		
Group	, armore _	Ind. %R2	Group %R2	Ind. %R2	Group %R2	Ind. %R2	Group %R2	Ind. %R2	Group %R2		
	Gender similarity	21,60	27,74	7,32	15,64	2,32	8,82	4,78	7,30		
Similarity	Ethnic similarity	0,93		2,34		1,15		1,69			
	Gender similarity & Ethnic similarity	5,21		5,99		5,35		0,84			
Entrepreneuer	At least one female proponent	-	-	-	-	15,41	15,41	5,95	5,95		
characteristics	At least one ethnic minority proponent	2,09	2,09	0,68	0,68	-	-	-	-		
C	Innovative	3,04	44,98	0,94	26,27	0,78	29,75	6,85	31,22		
Company characteristics	Firm age	8,12		20,03		20,90		12,55			
characteristics	Team size	33,82		5,30		8,07		11,82			
Campaign	Equity offering	0,39	6,31	0,34	0,89	0,24	1,21	0,46	3,17		
characteristics	Exit	5,92		0,55		0,96		2,71			
Investor characteristics	Serial investor	18,88	18,88	56,51	56,51	44,82	44,82	52,35	52,35		
					Owen valu	ue					
Group	Variable	Female - D	Oominant	Female -	Minority	Male -	Dominant	Male - Minority			
r	_	Ind. %R2	Group %R2	Ind. %R2	Group %R2	Ind. %R2	Group %R2	Ind. %R2	Group %R2		
	Gender similarity	28,73	42,12	5,35	30,07	6,04	16,29	14,19	19,98		
Similarity	Ethnic similarity	0,71		3,03		2,20		3,10			
	Gender & Ethnic similarity	12,68		21,69		8,04		2,69			
	Innovative	2,20	40,64	7,25	34,40	0,42	25,52	5,14	25,28		
Company characteristics	Firm age	8,86		2,95		21,29		14,19			
	Team size	29,58		24,20		3,81		5,94			
Campaign	Equity offering	0,84	6,24	0,17	4,61	0,43	1,26	0,60	2,62		
characteristics	Exit	5,40		4,44		0,83		2,02			
Investor characteristics	Serial investor	11,01	11,01	30,92	30,92	56,94	56,94	52,13	52,13		

Note: Under the general equation of (2), this table reports the fraction of the overall R-squared (per cent), explained by the different groups of characteristics – Owen value (Gr) - and the individual share (Ind) of the overall R-squared (per cent) associated with each variable belonging to a specific characteristic (Shapley decomposition). Note that these figures are median values within bootstrap percentile confidence intervals (level of confidence: 90%). This explains cases where Total is not always 100.00.

Results are showed for the sub-sample of female investors (Mod. 2) and male investors (Mod. 3), for the sub-sample of dominant investors (Mod. 4), for minority investors (Mod. 5) and the combination of gender and ethinicity of the investors: female investors that belong to the dominant ethnicity (Mod. 6), female investors that belong to the minority ethnicity (Mod. 7); male investors that belong to the dominant ethnicity (Mod. 8) and male investors that belong to the minority ethnicity (Mod. 9).

Year fixed effects are not included.

The following <u>company characteristics</u> are included in the regression: **Firm age** is the the numbers of years from the firm's establishment to the date of the campaign, **Innovative** is a dummy variable equals one in case of companies active in highly innovative sectors, **Team size** computed as the log of number of people listed on the site to make up the group of executive around the entrepreneur. Two <u>campaign structure characteristics</u> are included as follows: **Equity offering** is the percentage of equity offered during the campaign, **Exit** is a dummy variable equals 1 if the exit strategy is explicit in the campaign and zero otherwise. The variables related to the <u>entrepreneur characteristics</u> are the following: **At least one female proponent** is a dummy variable equals 1 in case of at least one female proponents among the overall numbers of proponents and zero otherwise, **At least one ethnic minority proponent** is a dummy variable equal to 1 for serial investors and zero for otherwise. <u>Similarity effects</u> rest on three variables: gender similarity effect - **Gender similarity** - is a dummy variable equals 1 in case of gender similarity between backers and proponents, ethnical similarity effect - **Ethnic similarity** - is a dummy variable equals 1 in case of gender similarity between backers and proponents.

 $Table\ 8-Similarity\ effect\ by\ gender\ and\ ethnicity\ groups\ of\ non-serial\ investors$ 

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Variables	All sample	Female	Male	Dominant	Minority	Female - Dominant	Female - Minority	Male - Dominant	Male - Minority
Gender similarity	-0.037	-0.801***	0.417***	-0.225**	0.330*	-0.821***	-1.114***	0.144	0.949***
	(0.092)	(0.199)	(0.113)	(0.111)	(0.171)	(0.240)	(0.362)	(0.135)	(0.210)
Ethnic similarity	-0.065	0.149	-0.150	-0.438***	0.757***	0.012	-0.204	-0.737***	1.479***
	(0.099)	(0.150)	(0.143)	(0.111)	(0.268)	(0.163)	(0.492)	(0.161)	(0.352)
Interaction [gender & ethnic	0.222*	0.049	0.294*	0.428***	-0.252	-0.218	3.302***	0.750***	-1.343***
similarity]	(0.114)	(0.257)	(0.157)	(0.131)	(0.326)	(0.296)	(0.758)	(0.181)	(0.413)
At least one female	-0.602***			-0.693***	-0.387**				
proponent	(0.076)			(0.084)	(0.181)				
At least one ethnic minority	0.377***	0.464**	0.332***						
proponent	(0.079)	(0.183)	(0.088)						
Innovative	-0.102	-0.064	-0.104	-0.072	-0.384*	-0.077	-0.282	-0.083	-0.395*
	(0.082)	(0.181)	(0.093)	(0.089)	(0.217)	(0.191)	(0.590)	(0.101)	(0.232)
Firm age	0.133***	0.068***	0.153***	0.139***	0.102***	0.079***	0.028	0.160***	0.131***
-	(0.012)	(0.025)	(0.014)	(0.013)	(0.030)	(0.026)	(0.066)	(0.015)	(0.034)
Team size	0.300***	0.181	0.357***	0.351***	0.128	0.277**	-0.055	0.440***	0.336**
	(0.056)	(0.132)	(0.064)	(0.060)	(0.149)	(0.140)	(0.359)	(0.071)	(0.166)
Equity offering	0.257	0.474	0.303	0.531	0.188	1.268	0.502	0.557	0.381
	(0.343)	(0.927)	(0.378)	(0.370)	(0.878)	(1.010)	(2.462)	(0.409)	(0.951)
Exit	0.785***	1.233***	0.719***	0.908***	-0.195	1.262***	-0.278	0.867***	-0.061
	(0.164)	(0.328)	(0.186)	(0.167)	(0.646)	(0.336)	(0.736)	(0.192)	(0.653)
Constant	5.513***	5.485***	5.010***	5.739***	5.458***	5.396***	5.544***	5.275***	4.659***
	(0.133)	(0.277)	(0.155)	(0.153)	(0.295)	(0.333)	(0.682)	(0.174)	(0.352)
Year fixed effect	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	6,074	1,381	4,693	5,000	1,074	1,164	217	3,836	857
R-squared	0.083	0.116	0.075	0.082	0.103	0.123	0.199	0.075	0.114
Adj. R-squared	0.0809	0.109	0.0731	0.0796	0.0926	0.115	0.160	0.0723	0.102

Notes: This table reports the results of a OLS regression. The dependent variable is the log of amount invested by the individual investor in a single campaign for the overall sample (Mod. 1), for the sub-sample of female investors (Mod. 2) and male investors (Mod. 3), for the sub-sample of dominant investors (Mod. 4), for minority investors (Mod. 5) and the combination of gender and ethinicity of the investors: female investors that belong to the dominant ethnicity (Mod. 6), female investors that belong to the minority ethnicity (Mod. 7); male investors that belong to the dominant ethnicity (Mod. 8) and male investors that belong to the minority ethnicity (Mod. 9). The following company features' controls are included in the regression: Firm age is the the numbers of years from the firm's establishment to the date of the campaign, Innovative is a dummy variable equals one in case of companies active in highly innovative sectors, Team size computed as the log of number of people listed on the site to make up the group of executive around the entrepreneur. Two campaign structure controls are included as follows: Equity offering is the percentage of equity offered during the campaign, Exit is a dummy variable equals 1 if the exit strategy is explicit in the campaign and zero otherwise. The control variables related to the entrepreneur characteristics are the following: At least one female proponent is a dummy variable equals 1 in case of at least one female proponents among the overall numbers of proponents and zero otherwise, At least one ethnic minority proponent is a dummy variable equals 1 in case of at least one ethnic minority proponents of proponents and zero otherwise. Variables related to the investor profile is Serial investor is a dummy variable equal to 1 for serial investors and zero for otherwise. Similarity effect rest on three variables: gender similarity effect - Gender similarity - is a dummy variable equals 1 in case of gender similarity between backers and proponents, ethnical similarity - is

Table 9 – Shapley and Owen value decomposition of R2 by non-serial investors characteristics

							Owen val	ue			
Group	Variable	All		Fen	nale	Male		Dominant		Minority	
Group	Variable	Ind. %R2	Group %R2	Ind. %R2	Group %R2	Ind. %R2	Group %R2	Ind. %R2	Group %R2	Ind. %R2	Group %R2
	Gender similarity	5.41	11.38	31.07	41.20	21.16	33.19	4.58	11.97	12.18	19.24
Similarity	Ethnic similarity	0.97		1.53		2.58		1.39		4.76	
	Gender similarity & Ethnic similarity	0.14		8.60		9.45		5.99		2.31	
	At least one female	30.26	31.45		3.04		0.87	33.83	33.83	15.25	15.25
Enrepreneuer characteristics	proponent At least one ethnic minority proponent	1.20		3.04		0.87	0.07	55.65	22.62	10.20	10.20
	Innovative	3.29	54.78	4.27	49.09	2.76	63.13	2.52	51.56	9.70	62.08
Company characteristics	Firm age	35.42		8.83		49.99		36.47		25.39	
	Team size	16.06		35.99		10.38		12.56		26.99	
Campaign	Equity offering	0.42	2.39	0.37	6.66	0.84	2.80	0.47	2.63	0.46	3.42
characteristics	Exit	1.97		6.29		1.95		2.16		2.96	
							Owen val	ue			
Group	Variable			Female - l	Dominant	Female - Minority		Male - Dominant		Male - Minority	
Group	Variable			Ind. %R2	Group %R2	Ind. %R2	Group %R2	Ind. %R2	Group %R2	Ind. %R2	Group %R2
	Gender similarity			35.09	54.04	9.30	63.19	16.40	34.86	34.38	51.58
Similarity	Ethnic similarity			0.96		5.59		4.59		9.19	
	Gender & Ethnic similarity			17.97		48.30		13.87		8.01	
	Innovative			3.66	40.01	4.97	30.15	1.99	61.17	7.13	46.61
Company characteristics	Firm age			8.16		3.99		52.31		26.54	
characteristics	Team size			28.19		21.19		6.86		12.94	
Campaign	Equity offering			0.65	5.95	0.22	6.65	1.14	3.97	0.43	1.81
characteristics	Exit			5.30		6.43		2.83		1.38	

Note: Under the general equation of (2), this table reports the fraction of the overall R-squared (per cent), explained by the different groups of characteristics – Owen value (Gr) - and the individual share (Ind) of the overall R-squared (per cent) associated with each variable belonging to a specific characteristic (Shapley decomposition). Note that these figures are median values within bootstrap percentile confidence intervals (level of confidence: 90%). This explains cases where Total is not always 100.

# CEFIN Working Papers ISSN (online) 2282-8168

- 79 The effect of the Fed zero-lower bound announcement on bank profitability and diversification, by Landi, A. Sclip, A. and Venturelli, V. (December 2019).
- 78 Portfolio choice: Evidence from new-borns, by Arnaboldi, F. and Gioia, F. (October 2019).
- 77 Behind the success of dominated personal pension plans: sales force and financial literacy factors, by Marotta, G. (October 2019).
- Short-term household income mobility before and after the Great Recession: A four country study, by Casabianca, E.J. and Giarda, E. (June 2019).
- 75 The impact of Fundamental Review of Trading Book: A preliminary assessment on a stylized portfolio, by Pederzoli, C. and Torricelli, C. (February 2019).
- Security-voting structure and equity financing in the Banking Sector: 'One Head-One Vote' versus 'One Share-One Vote', by Ferretti, R., Pattitoni, P. and Castelli, A. (January 2019).
- 73 The Financial Decisions of Immigrant and Native Households: Evidence from Italy, by Bertocchi, G., Brunetti, M. and Zaiceva, A. (January 2019).
- Why choosing dominated personal pension plans: sales force and financial literacy effects, by Marotta, G. (July 2018).
- 71 Spending Policies of Italian Banking Foundations, by Pattarin, F. (July, 2018).
- 70 The Forecasting Performance of Dynamic Factors Models with Vintage Data, by Di Bonaventura, L., Forni, M. and Pattarin, F. (July 2018).
- 69 Housing Market Shocks in Italy: a GVAR Approach, by Cipollini, A. and Parla, F. (April, 2018).
- 68 Customer Complaining and Probability of Default in Consumer Credit, by Cosma, S., Pancotto, F. and Vezzani, P. (March, 2018).
- 67 Is Equity Crowdfunding a Good Tool for Social Enterprises?, by Cosma, S., Grasso, A.G., Pagliacci, F. and Pedrazzoli, A. (February, 2018).
- 66 Household Preferences for Socially Responsible Investments, by Rossi. M.C., Sansone, D., Torricelli, C. and van Soest, A. (February2018).
- 65 Market-Book Ratios of European Banks: What does Explain the Structural Fall?, by Ferretti, R.,Gallo, G., Landi, A. and Venturelli. V. (January 2018).
- 64 Past Income Scarcity and Current Perception of Financial Fragility, by Baldini, M., Gallo, G. and Torricelli, C. (December 2017).
- 63 How does Financial Market Evaluate Business Models? Evidence from European Banks, by Cosma, S., Ferretti, R., Gualandri, E., Landi, A. and Venturelli, V. (May 2017)
- 62 Individual Heterogeneity and Pension Choices. How to Communicate an Effective Message?, by Gallo, G., Torricelli, C. and van Soest, A. (April 2017)
- 61 The Risk-Asymmetry Index, by Elyasiani, E., Gambarelli L. and Muzzioli, S.(December 2016)
- Strumenti per il sostegno finanziario di famiglie e microimprese: il caso italiano (Policies to help financially vulnerable Italian households and micro-businesses), by Cotterli, S. (December 2016)
- A test of the Behavioral versus the Rational model of Persuasion in Financial Advertising, by Ferretti, R., Pancotto, F. and Rubaltelli, E. (May 2016)
- 58 Financial connectedness among European volatility risk premia, by Cipollini, A., Lo Cascio, I. and Muzzioli, S. (December 2015)
- 57 The effectiveness of insider trading regulations. The case of the Italian tender offers, by Ferretti, R., Pattitoni, P. and Salinas, A. (September 2015)
- Introducing Aggregate Return on Investment as a Solution for the Contradiction between some PME Metrics and IRR, by Altshuler D. and Magni, C.A. (September 2015)
- 55 'It's a trap!' The degree of poverty persistence in Italy and Europe, by Giarda, E. and Moroni, G. (September 2015)
- 54 Systemic risk measures and macroprudential stress tests. An assessment over the 2014

- EBA exercise. by Pederzoli, C. and Torricelli, C. (July 2015)
- Emotional intelligence and risk taking in investment decision-making, by Rubaltelli, E., Agnoli, S., Rancan, M.and Pozzoli, T. (July 2015)
- 52 Second homes: households' life dream or wrong investment?, by Brunetti, M. and Torricelli, C. (May 2015)
- Pseudo-naïve approaches to investment performance measurement, by Magni, C.A. (February, 2015)
- Monitoring systemic risk. A survey of the available macroprudential Toolkit, by Gualandri, E. and Noera, M. (November 2014).
- Towards a macroprudential policy in the EU: Main issues, by Gualandri, E. and Noera, M. (November 2014).
- Does homeownership partly explain low participation in supplementary pension schemes?, by Santantonio, M., Torricelli, C., and Urzì Brancati M.C., (September 2014)
- 47 An average-based accounting approach to capital asset investments: The case of project finance, by Magni, C.A. (September 2014)
- 46 Should football coaches wear a suit? The impact of skill and management structure on Serie A Clubs' performance, by Torricelli, C., Urzì Brancati M.C., and Mirtoleni, L. (July 2014)
- 45 Family ties: occupational responses to cope with a household income shock, by Baldini, M., Torricelli, C., Urzì Brancati M.C. (April 2014)
- 44 Volatility co-movements: a time scale decomposition analysis, by Cipollini, I., Lo Cascio I., Muzzioli. S. (November 2013)
- The effect of revenue and geographic diversification on bank performance, by Brighi, P., Venturelli, V. (October 2013)
- The sovereign debt crisis: the impact on the intermediation model of Italian banks, by Cosma,S., Gualandri, E. (October 2013)
- The financing of Italian firms and the credit crunch: findings and exit strategies, by Gualandri, E., Venturelli, V. (October 2013)
- 40 Efficiency and unbiasedness of corn futures markets: New evidence across the financial crisis, by Pederzoli, C., Torricelli, C. (October 2013)
- 39 La regolamentazione dello short selling: effetti sul mercato azionario italiano (Short selling ban: effects on the Italian stock market), by Mattioli L., Ferretti R. (August 2013)
- A liquidity risk index as a regulatory tool for systematically important banks? An empirical assessment across two financial crises, by Gianfelice G., Marotta G., Torricelli C. (July 2013)
- 37 Per un accesso sostenibile delle Pmi al credito (A sustainable access to credit for SMEs), by Marotta, G. (May 2013)
- The unavoidable persistence of forum shopping in the Insolvency Regulation, by Mucciarelli, F.M. (April 2013)
- Rating Triggers, Market Risk and the Need for More Regulation, by Parmeggiani, F. (December 2012)
- 34 Collateral Requirements of SMEs: The Evidence from Less–Developed Countries, by Hanedar, E.H. Broccardo, E. and Bazzana, F. (November 2012)
- 33 Is it money or brains? The determinants of intra-family decision power, by Bertocchi, G., Brunetti, M. and Torricelli, C. (June 2012)
- 32 Is financial fragility a matter of illiquidity? An appraisal for Italian households, by Brunetti, M, Giarda, E. and Torricelli, C. (June 2012)

- 31 Attitudes, personality factors and household debt decisions: A study of consumer credit, by Stefano Cosma and Francesco Pattarin (February 2012)
- 30 Corridor implied volatility and the variance risk premium in the Italian market, by Silvia Muzzioli (November 2011)
- 29 Internal Corporate Governance and the Financial Crisis: Lessons for Banks, Regulators and Supervisors, by Gualandri, E., Stanziale, A. and Mangone, E. (November 2011)
- Are defined contribution pension schemes socially sustainable? A conceptual map from a macroprudential perspective, by Marotta, G. (October 2011)
- 27 Basel 3, Pillar 2: the role of banks' internal governance and control function, by Gualandri, E. (September 2011)
- 26 Underpricing, wealth loss for pre-existing shareholders and the cost of going public: the role of private equity backing in Italian IPOs, by Ferretti, R. and Meles, A. (April 2011)
- 25 Modelling credit risk for innovative firms: the role of innovation measures, by Pederzoli C., Thoma G., Torricelli C. (March 2011)
- 24 Market Reaction to Second-Hand News: Attention Grabbing or Information Dissemination?, by Cervellati E.M., Ferretti R., Pattitoni P. (January 2011)
- Towards a volatility index for the Italian stock market, by Muzzioli S. (September 2010)
- 22 A parsimonious default prediction model for Italian SMEs, by Pederzoli C., Torricelli C. (June 2010)
- 21 Average Internal Rate of Return and investment decisions: a new perspective, by Magni C.A. (February 2010)
- The skew pattern of implied volatility in the DAX index options market, by Muzzioli S. (December 2009)
- Accounting and economic measures: An integrated theory of capital budgeting, by Magni C.A. (December 2009)
- 18 Exclusions of US-holders in cross-border takeover bids and the principle of equality in tender offers, by Mucciarelli F. (May 2009).
- 17 Models for household portfolios and life-cycle allocations in the presence of labour income and longevity risk, by Torricelli C. (March 2009)
- Differential evolution of combinatorial search for constrained index tracking, by Paterlini S, Krink T, Mittnik S. (March 2009)
- Optimization heuristics for determining internal rating grading scales, by Paterlini S, Lyraa M, Pahaa J, Winker P. (March 2009)
- The impact of bank concentration on financial distress: the case of the European banking system, by Fiordelisi F, Cipollini A. (February 2009)
- 13 Financial crisis and new dimensions of liquidity risk: rethinking prudential regulation and supervision, by Landi A, Gualandri E, Venturelli V. (January 2009)
- 12 Lending interest rate pass-through in the euro area: a data-driven tale, by Marotta G. (October 2008)
- Option based forecast of volatility: an empirical study in the Dax index options market, Muzzioli S. (May 2008)
- 10 Lending interest rate pass-through in the euro area, by Marotta G. (March 2008)

- 9 Indebtedness, macroeconomic conditions and banks' losses: evidence from Italy, by Torricelli C, Castellani S, Pederzoli C. (January 2008)
- 8 Is public information really public? The role of newspapers, Ferretti R, Pattarin F. (January 2008)
- 7 Differential evolution of multi-objective portfolio optimization, by Paterlini S, Krink T. (January 2008)
- 6 Assessing and measuring the equity gap and the equity, by Gualandri E, Venturelli V. (January 2008)
- Model risk e tecniche per il controllo dei market parameter, Torricelli C, Bonollo M, Morandi D, Pederzoli C. (October 2007)
- The relations between implied and realised volatility, are call options more informative than put options? Evidence from the Dax index options market, by Muzzioli S. (October 2007)
- 3 The maximum LG-likelihood method: an application to extreme quantile estimation in finance, by Ferrari D., Paterlini S. (June 2007)
- 2 Default risk: Poisson mixture and the business cycle, by Pederzoli C. (June 2007)
- 1 Population ageing, household portfolios and financial asset returns: a survey of the literature, by Brunetti M. (May 2007)